



Limnology and Oceanography Letters 7, 2022, 251–260 © 2021 The Authors. Limnology and Oceanography Letters published by Wiley Periodicals LLC on behalf of Association for the Sciences of Limnology and Oceanography. doi: 10.1002/10/21.0225

### LETTER

# Thermal sensitivity in dual-breathing ectotherms: Embryos and mothers determine species' vulnerability to climate change

Lyle D. Vorsatz <sup>(1)</sup>, <sup>1</sup> Bruce P. Mostert,<sup>2</sup> Christopher D. McQuaid <sup>(1)</sup>,<sup>2</sup> Stefano Cannicci <sup>(1)</sup>,<sup>1,3</sup>\* Francesca Porri <sup>(1)</sup>,<sup>2,4</sup>

<sup>1</sup>The Swire Institute of Marine Science and the School of Biological Sciences, The University of Hong Kong, Hong Kong, Hong Kong SAR; <sup>2</sup>Department of Zoology & Entomology, Rhodes University, Grahamstown, South Africa; <sup>3</sup>Department of Biology, University of Florence, Sesto Fiorentino, Italy; <sup>4</sup>South African Institute for Aquatic Biodiversity, Grahamstown, South Africa

### Scientific Significance Statement

The thermal bottlenecks and the response to climate change in the complex life-history of intertidal ectotherms are not clearly defined. Our work tested the applicability of two major ecological and mechanistic hypotheses across life stages, latitudinal range and between respiratory media. We found that ontogenetic stage and local climate adaptations are significant factors to consider when evaluating the overall vulnerability of intertidal ectotherms to thermal stress. The evolutionary adaptation and capacity for dual breathing acquired throughout ontogeny by some intertidal ectotherms may represent an advantage in mitigating the effects of climate change in comparison to strictly water breathers.

### **Abstract**

Understanding the life-stage specific vulnerability of ectotherms to temperature increases is crucial to accurately predicting the consequences of current and future global climate change. Here, we examined ontogeny-specific thermal vulnerability of three intertidal, bimodal (i.e., air and water) breathing crabs from tropical and warm temperate latitudes to address this issue. Spawning females and embryos of intertidal crabs from warm temperate latitudes were more vulnerable to temperature increases than tropical conspecifics, particularly in water. Our findings do not fully support the Climate Variability Hypothesis for setting upper thermal limits, but correspond with the Oxygen- and Capacity-Limited Thermal Tolerance hypothesis, suggesting ontogeny-specific aerobic capacity dictates overall species' thermal sensitivity. Bimodal breathing efficiency as an evolutionary adaptation, ontogenetic stage and local climate adaptation are therefore significant factors to consider when evaluating the vulnerability of intertidal ectotherms to temperature increases and the consequences of climate changes for intertidal organisms, populations and communities.

Data Availability Statement: Data and metadata are available on the Open Science Framework at: https://osf.io/zpkfn/?view\_only= ad146b74b8464e4abae97596e6688b1e.

### Associate editor: Rana El-Sabaawi

Additional Supporting Information may be found in the online version of this article.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

<sup>\*</sup>Correspondence: cannicci@hku.hk

**Author contribution statement:** LDV conducted the analyses and wrote the first draft of the manuscript. BPM, FP, CDM and SC conceived the study. BPM, CDM, SC and FP provided data and expert guidance. FP, CDM and SC provided the funding. All authors contributed to writing and editing the manuscript.

The Oxygen- and Capacity-Limited Thermal Tolerance (OCLTT) hypothesis and the Climate Variability Hypothesis (CVH) are strongly supported mechanistic and conceptually unifying principles explaining the onset of metabolic stress and vulnerability to temperature increases along latitudinal gradients (Huey and Kingsolver 1989; Stevens 1989; Frederich and Pörtner 2000; Pörtner and Farrell 2008; Sunday et al. 2019). The OCLTT proposes that during short- or longterm thermal stress at the limits of an organism's thermal window (the temperature range between an organism's lower and upper thermal limits), responses in the respiration and circulatory systems fulfill the increased oxygen demands. When these systems can no longer cope with the increased demand, however, the organism's thermal tolerance is constrained by the onset of lower than normal levels of oxygen in their tissues (Frederich and Pörtner 2000; Pörtner and Farrell 2008; Verberk et al. 2016). The OCLTT predicts that the effects of thermal stress are alleviated by an increased supply of oxygen either by optimization of respiratory mechanisms or by an increased supply from the environment, thus widening the thermal tolerance window (Frederich and Pörtner 2000). Meanwhile, the CVH proposes that organisms which experience small variations of temperature in their environments are expected to have narrower thermal windows and higher upper thermal limits than those living in highly variable environments (Sunday et al. 2011; Shah et al. 2017). The OCLTT has been widely investigated (for a review see Verberk et al. 2016) and has been found to apply to aquatic rather than terrestrial animals, which is expected, due to the relationship between oxygen solubility and temperature, which strongly differs in air and water. Meanwhile, the CVH has been shown to apply to ectotherms in both the marine and terrestrial realms, but still needs validation for intertidal ectotherms that can respire in air and water (Sunday et al. 2019).

Traditionally, climate change vulnerability has been evaluated through active metabolic responses to temperature and critical thermal limits (Deutsch et al. 2008; Pinsky et al. 2019). Notwithstanding their use, these approaches overlook the sublethal effects of temperature on energy budgets and the critical limits of performance traits, where increases and peaks in oxygen consumption represent thermal optima (Kellermann et al. 2019). This is, however, not the case for standard metabolic rate, where peaks in the thermal performance curve (TPC), which describe the effects of temperature on biological process rates, represent elevated minimum costs, typically at higher temperatures than those produced by other performance traits (Kellermann et al. 2019; Shah et al. 2021). A steep increase in standard metabolic rate toward a peak is therefore considered as the onset of stress, which affects the aerobic scope for activity, that is, the energy available for other metabolically costly activities such as locomotion, growth, and reproduction (Pörtner 2001). Adaptations to local climate and dual respiratory media of some intertidal ectotherms likely involve trade-offs in their energy budgets

at different ontogenetic stages and are likely to differ from those that are exclusively water breathers.

To date, investigations into the thermal dependence of intertidal ectotherms that encompass all life stages from embryo to reproductive adults are lacking. This suggests that the potentially most thermally vulnerable stages within and among the life cycles of a large representative group of ectotherms are still unknown or continuously missed due to a lack of experimental data. Moreover, the life stages most vulnerable to temperature increases across a latitudinal gradient in different respiratory media (water or air) could result in critically different outcomes due to local thermal adaptation and species-specific cardiorespiratory functioning. Here, we address these interconnected issues by examining the stage-specific (embryo, larva, adult, and brooding adult) thermal vulnerability of three intertidal, bimodal breathing ectotherms, the mangrove crabs Tubuca urvillei and the congeneric Parasesarma guttatum and Parasesarma capensis, as model organisms from tropical and warm temperate regions.

### Methods and materials

### Model species and study area

Parasesarma guttatum and P. capensis are congeners endemic to the Western Indian Ocean where the distribution of the latter extends from the southernmost mangroves in South Africa to the Mozambican channel. Farther north, P. capensis is replaced by P. guttatum, which extends north to Somalia (Fratini et al. 2019). The distribution of the fiddler crab T. urvillei extends from South Africa to Somalia along the coasts of east African and Madagascar (Shih et al. 2018). All three species are considered to be keystone taxa in mangroves due to their bioengineering capabilities, contributing to both nutrient and chemical cycling (Cannicci et al. 2008). Two study areas were selected to represent a tropical, low latitude region (Gazi Bay, Kenya; 4°22'S, 39°30'E) and a warm-temperate, higher latitude region (Mngazana, South Africa; 31°42"S, 29°25"E; Fig. 1). See Appendix S1, SI 1 for more details on site descriptions and environmental characterization.

## Animal collection, maintenance, and general experimental design

Gravid and non-gravid female *P. guttatum, P. capensis* and *T. urvillei* were collected by hand in Kenya and South Africa. Both non-gravid and gravid crabs brooding stage two and four embryos (Simoni et al. 2011) were maintained in individual plastic containers tilted at a  $45^{\circ}$  angle partially filled with locally collected mangrove mud and freshly aerated seawater. Females brooding stage five embryos (Simoni et al. 2011) were placed in separate holding tanks with rocks and aerated seawater. Larvae were successfully hatched from stage five embryos within 2 days of collection for ensuing experiments. All life stages were subjected to either an increasing/decreasing thermal ramp (the rate of temperature change over an interval) starting at  $27^{\circ}$ C up to  $35^{\circ}$ C or  $27^{\circ}$ C down to  $19^{\circ}$ C at a



Fig. 1. Map of the latitudinal ranges of the model species (based on Shih et al. 2016; Fratini et al. 2019). Stars represent the sampling areas; double pink and pale blue lines represent *Tubuca urvillei*; dark blue line represents *Parasesarma capensis*; orange line represents *Parasesarma guttatum*.

rate of 1°C h<sup>-1</sup>, respectively (Fusi et al. 2015). Using an intermittent flow-through respirometry technique, oxygen consumption was recorded every 2°C along the thermal ramp until at least a 5% decrease in oxygen (µmol) content of the test medium was detected. Oxygen saturation was never allowed to fall below 60% to avoid possible hypoxic effects (Schurmann and Steffensen 1992). Additionally, during each trial, a vial/chamber/syringe was left empty for aerial experiments or filled with aerated seawater to control for background respiration and oxygen consumption rates  $(MO_2)$ adjusted accordingly. New animals were used for each trial along each direction of the thermal ramp. Prior to the start of each experiment in both regions and media, all oxygen sensor types were two-point calibrated in oxygen-free and airsaturated seawater, the former containing a solution of 1% sodium dithionite (Na<sub>2</sub>S<sub>2</sub>O<sub>4</sub>). To ensure adequate mixing of water for all systems; the syringes were gently inverted

throughout trials and electro-magnetic stirrers were used for 0.5- and 2-mL vials. A minimum of eight replicates at each experimental temperature was tested for each embryonic and larval stage in both regions and respiratory media. Replicates involved individual animals except in the case of embryos and larvae; for these, replicates involved multiple individuals due to their small size. For each technical replicate, embryos and hatched larvae from the same female were used. Upon trial completion (which typically lasted 1–2 h), embryos and larvae were removed, counted and standardized with  $MO_2$  expressed as nmol  $O_2$  individual<sup>-1</sup> min<sup>-1</sup>. For more life-stage-specific experimental design details, see Appendix S1.

### Data analysis

All analyses were conducted in R for computing statistics (R core team 2019). Temperature and  $MO_2$  were assessed for normality and homoscedasticity. Natural variability in

temperature was examined using an Aligned Rank Transformation ANOVA (ARTANOVA) of the temperature data to test for differences among positions in relation to the sediment (20 cm below, 1 cm and 1.5 m above), between regions and the interaction between these two independent (fixed) variables using the "ARTool" package (Wobbrock et al. 2011). Gaussian TPCs were calculated for each of the model species, ontogenetic stages, regions, media and reproductive states and activation energies (Ea) extracted using the "calc\_params" function in the R packages "nls.multstart" and "rTPC" (Padfield et al. 2021). A generalized linear mixed model (GLMM) was conducted using the "lme4" package (Bates et al. 2015) to test for differences in metabolic rate, where  $MO_2$  was the response variable and temperature, ontogenetic stage (stage two embryos/ stage four embryos/ zoeae I), medium (water/air) and species partially nested in region were the independent fixed variables, with sample ID as a random variable (Schielzeth and Nakagawa 2013). Separate GLMM's for each ontogenetic stage were also conducted using MO<sub>2</sub> as the response variable and temperature, medium (water/air) and species nested in region as independent variables to identify stage-specific patterns among media, regions and species. To test MO<sub>2</sub> as a proxy for the maternal effects of brooding embryos, GLMM's were conducted using temperature, medium, reproductive state (gravid/non-gravid) and species partially nested in region as independent variables with sample ID as a random variable (Schielzeth and Nakagawa 2013). Where applicable, all post hoc tests were conducted with a Benjamini-Hochberg correction (Benjamini and Hochberg 1995) using the "emmeans" package. Particulars for the analysis of life-stage-specific thermal responsiveness using the Arrhenius-Boltzmann model (Clarke 2017) are detailed in Appendix S1, SI 1.

### Results

There were significant differences in the temperature patterns observed in each region: Kenya and South Africa (ARTANOVA:  $F_{1,38,876} = 22,415.76$ , p > 0.001), position in the sediment: 20 cm below, 1 cm and 1.5 m above (ARTANOVA:  $F_{1,38,876} = 1238.39$ , p > 0.001) and the interaction between region and position (ARTANOVA:  $F_{1,38,876} = 277.34$ , p > 0.001; Fig. S1). Furthermore, all post hoc pairwise comparisons of temperature were significantly different with the exception of measurements taken 1.5 m above and 1 cm above the sediment surface in South Africa (Table S1).

### **Oxygen consumption**

The metabolic responses to increasing temperature differed significantly across regions, species, ontogenetic stages and respiratory media (Fig. 2, Table S2). Metabolic rates significantly differed across species, regions and respiratory media for stage two embryos (Table 1), stage four embryos (Table 1)

and stage I zoeae (Table 2). At stage two and four of embryo development of all species examined, oxygen consumption was generally significantly greater in water than air (Figs. 2A, B. Tables S3 and S4). There were no significant differences in the MO<sub>2</sub> for stage two embryos among species and regions when tested in air (Fig. 2A, Table S3). Differences among species and between regions however occurred when tested in water for both stage two and four embryos (Figs. 2A,B. Tables S3 and S4). Furthermore, under increasing temperatures, the early ontogenetic stages (stage two and stage four embryos and stage I zoeae) of Kenyan populations of T. urvillei and P. guttatum exhibited significantly lower MO<sub>2</sub> in water than those of South African populations of T. urvillei and P. capensis. The South African populations of stage two P. capensis and stage four T. urvillei embryos showed signs of the onset of stress at temperatures above  $\sim 31^{\circ}$ C in water, indicating a critical thermal threshold at around  $\sim 29^{\circ}$ C (Fig. 2A, B). The Kenvan populations exhibited no such onset of thermal stress, with metabolic rate increasing gradually with temperature in both air and water for both embryos stages and zoeae. The Kenyan populations of P. guttatum and T. urvillei stage I zoeae showed no significant differences in  $MO_{2}$ . whereas the South African populations of T. urvillei and P. capensis MO<sub>2</sub> were significantly higher than the Kenyan P. guttatum and T. urvillei (Fig. 2C; Table S5).

The  $MO_2$  of adult crabs differed significantly between regions, species, reproductive states and respiratory media (Fig. 2D,E, Table S6). *T. urvillei* showed no significant differences between regions or reproductive states in either air or water (Table S7). In contrast, gravid females of *P. capensis* and *P. guttatum* exhibited significantly higher  $MO_2$  than nongravid females in both air and water, with *P. capensis* showing a marked increase in the difference between the reproductive states at temperatures above ~29°C in air and 27°C in water (Fig. 2D,E). Furthermore, in both air and water, gravid *P. capensis* showed a plateau in metabolism above 30°C, indicating that the onset of stress occurred between 27°C and 29°C before metabolic rate decreased as oxygen became limiting for the upkeep of basal metabolism (Fig. 2D,E).

Direct comparisons of thermal responsiveness revealed that slopes from the Boltzmann–Arrhenius model differed significantly for each life-stage, exemplified by the significant effects of the interaction terms in the linear mixed models for stage two and four embryos (Tables S8 and S9), stage one zoeae (Table S10), and female adult crabs (Table S11). The early ontogenetic stages (stage two and stage four embryos and stage I zoeae) of Kenyan populations of *T. urvillei* and *P. guttatum* had lower thermal responsiveness than the South African populations of *T. urvillei* and *P. capensis*, particularly in water (Fig. S2A–C; Tables S13–S16). While, gravid *P. capensis* from temperate latitudes were more thermally responsive than non-gravid females in both air and water (Fig. S2D,E; Table S17).



**Fig. 2.** Thermal performance curves of *Tubuca urvillei, Parasesarma capensis* and *Parasesarma guttatum* from Kenya (low latitude/tropical) and South Africa (higher latitude/warm temperate) in air and water across (A) stage two embryos, (B) stage four embryos, (C) zoeae I (in water only), (D) gravid and non-gravid females in air and (E) gravid and non-gravid females in water. Thermal performance curves for populations based on metabolic rate ( $MO_2$  as a proxy) with 95% bootstrapped confidence intervals. Color coded points represent raw  $MO_2$  per individual at a given temperature. \* No measurements were recorded for gravid and non-gravid adults between 19 and 25°C in Kenya (D and E).

**Table 1.** Generalized linear mixed effects model (GLMM) evaluating the effect of temperature, respiratory media (water vs. air), species partially nested in region (South Africa vs. Kenya) and their interactions on the oxygen consumption ( $MO_2$ ) of stage two and stage four embryos. SS, sum of squares; DF, degrees of freedom; F, Fisher's statistic; *p*, significance value. All significant results (*p* < 0.05) are indicated in bold. \*\**p* < 0.01; \**p* < 0.05.

Effect	DF	Stage two		Stage four	
		SS	F	SS	F
Temperature	1	0.008	242.032**	0.007	115.792**
Species	2	0.002	28.502**	0.001	8.809**
Respiratory medium	1	0.003	106.545**	0.003	48.504**
Temperature $\times$ species	2	0.003	53.313**	0.002	15.908**
Temperature $\times$ respiratory medium	1	0.006	196.079**	0.006	105.952**
Species $\times$ respiratory medium	2	0.002	41.812**	0.009	76.793**
Temperature $\times$ species $\times$ region	1	0.003	82.342**	0.001	17.355**
Temperature $\times$ species $\times$ respiratory medium	2	0.004	63.936**	0.009	76.314**
Temperature × species × region × respiratory medium	1	0.002	1.514	0.001	5.381*

**Table 2.** Generalized linear mixed effects model (GLMM) evaluating the effect of temperature, species partially nested in region and their interactions on the oxygen consumption ( $MO_2$ ) of stage I zoeae in water only. SS, sum of squares; DF, degrees of freedom; F, Fisher's statistic; *p*, significance value. All significant results (*p* < 0.05) are indicated in bold.

Effect	DF	SS	F	р
Temperature	1	0.686	159.617	< 0.001
Species	2	0.351	40.816	< 0.001
Temperature $\times$ species	2	0.599	69.719	< 0.001
Temperature $\times$ species $\times$ region	1	0.221	51.288	< 0.001

### Discussion

Here, we tested the ontogeny-specific responses to temperature increases among intertidal, bimodal breathing model organisms from tropical and warm temperate regions. Our results indicate that at warm temperate latitudes, embryos are the most responsive and vulnerable to temperature increases, regardless of medium, but particularly in water. Additionally, the energetic cost of carrying and brooding embryos potentially reduces performance and, possibly, survivorship in females, again regardless of respiratory medium. These findings are consistent with the OCLTT principle, which postulates that organisms in aquatic habitats are more constrained in supplying oxygen to their tissues, and thus more prone to loss of performance than animals breathing air. This theory also suggests that aerobic capacity changes throughout ontogeny (Frederich and Pörtner 2000; Pörtner and Farrell 2008; Verberk et al. 2016). Moreover, our results indicate that dual breathing intertidal ectotherms living at warm temperate latitudes are at greater risk to temperature increases than those in the tropics (Huey and Kingsolver 1989; Stevens 1989; Sunday et al. 2012). This does not fully support the predictions of the

CVH, which suggests the upper thermal limits of species are set according to temperature variability as linked to latitude, but the CVH may still be useful for predicting the width of dual breathing intertidal ectotherm thermal windows.

Stage-specific aerobic performance of decapod crustaceans reflects the development of their osmoregulatory and cardiorespiratory systems (Charmantier et al. 2002; Small et al. 2015). Embryos, particularly in aquatic habitats, are the least efficient at supplying oxygen to their tissues in the life cycle of ectotherms and have the least developed homeostatic functions (Fernandez et al. 2003; Hamdoun and Epel 2007). As shown here, warm temperate embryos were more thermally responsive and exhibited signs of oxygen limitation in water; rising temperatures resulted in an initial increase in metabolic rate, followed by an asymptote or a decline. With a high demand for oxygen, the egg membrane acts as a barrier to diffusion, potentially limiting oxygen uptake and delaying hatching (Fernandez et al. 2003). This in turn reduces the fitness of the hatched larva and causes catch-up growth, i.e., a metabolically costly convergence of growth trajectories after initial stress (Anger 2001). An immature homeostatic and aerobic capacity in marine-dependent ectotherms exacerbates their sensitivity to temperature, salinity, hypoxia and other environmental stressors, posing a potential bottleneck for survivorship and successful development under conditions of climate change (Przeslawski et al. 2015).

Brooding females invest significant amounts of energy in oogenesis, producing large quantities of embryos that increase their metabolic load, leading to high energetic costs (Hartnoll 2006). These costs hinder the maintenance of standard metabolic functions due to the energy spent on reproduction in addition to the increase in body mass (due to carrying an egg mass) of brooding females and thus may reduce their aerobic capacity (Pörtner and Farrell 2008; Sokolova 2013). Similar observations have recently been reported for fish, in which spawning adults and fish embryos consistently present narrower thermal tolerances than larvae and nonreproductive adults, indicating that these critical stages in their life cycles can form population bottlenecks (Dahlke et al. 2020).

Our model organisms showed distinct adaptations to the local climatic conditions of each region. When tested in water, those collected at temperate latitudes exhibited higher oxygen consumption and greater thermal responsiveness at all early ontogenetic stages than those from the tropics. Here, we show that ectotherms from temperate latitudes are more vulnerable to acute heating events at all life stages, showing greater thermal responsiveness, most critically when brooding embryos in the warmer months. This vulnerability is probably due to metabolic upregulation from eurythermal adaptation that elevates standard organismal oxygen demand when compared to warmer acclimated (tropical) ectotherms at the same temperatures (Pörtner and Gutt 2016). This is a cause for concern as extreme temperature events are forecast to occur with more frequency and intensity at mid- than at low latitudes (IPCC 2014; Hayashida et al. 2020; Oliver et al. 2021). If these events coincide with the critical bottleneck stages outlined here, local extinction events are likely to occur due to the limited thermal plasticity of ectotherms (Gunderson and



**Fig. 3.** Conceptual model adapted from Pörtner and Farrell (2008) illustrating the thermal sensitivity of (A) higher latitude/warm temperate and (B) low latitude/tropical brachyuran bimodal breathers throughout ontogeny. Note megalopae and zoeae only occur in water. Temperature-dependent aerobic metabolic performance is maintained throughout the local temperature range (horizontal bar). In both tropical and higher-latitude adapted bimodal breathers in water, the life cycle stages most sensitive to acute heat events are the embryo and brooding female. Tropical and higher-latitude adapted bimodal breathers are not as constrained in air and thus have a higher tolerance to acute heat events. Tropical adapted bimodal breathers in both respiratory mediums have a higher tolerance to increases in temperature than higher latitude/warm temperate adapted populations and congeners.

Vorsatz et al.

Stillman 2015). Decapods in the tropics have likely evolved adaptive metabolic compensation that regulates oxygen consumption at extreme upper temperatures. This will have been strongly driven by the experienced year-round climate as a mechanism to increase heat tolerance, as indicated by their lower activation energies (Frederich and Pörtner 2000). Our data indicate that bimodal breathing intertidal ectotherms in the tropics are likely to be better able to endure acute heat events, with all early life stages able to acquire sufficient oxygen with no signs of metabolic stress. This mirrors the findings of Fusi et al. (2015) on the thermal sensitivity of adult males of the same model species from the same regions. The pervasive nature of this sensitivity, from embryos through to adults, further underlines the thermal risks that higher latitude, temperate ectotherm populations are, and will be, exposed to. Thus, generalizations predicting thermal vulnerability of ectotherms along a latitudinal gradient should be made more cautiously. As shown here, such predictions may not be universally applicable and multilevel interpretation must be considered, especially for organisms already living in thermally stressful environments. Our focus on dual-breathing intertidal ectotherms like crabs, rather than the more frequently studied exclusively water-breathing animals, fills an important gap in the link between physiological vulnerability and climate change (Huey et al. 2012).

Based on our results, we propose a conceptual model to highlight the differences in thermal sensitivity among life stages, geographical regions and breathing media for intertidal ectotherms (Fig. 3). The megalopal response to temperature increases as depicted in in the model is extrapolated from Vorsatz et al. (2021) for *Parasesarma catenatum* and *Neosarmatium africanum*. The juvenile response to temperature increases are extrapolated from Walther et al. (2010) for the spider crab *Hyas araneus* assuming that juveniles of brachyurans crabs regardless of primary breathing mode have a higher thermal tolerance than megalopae, but a lower thermal tolerance than adults.

The geographic-, ontogenetic-, and respiratory mediumdependent thermal response patterns unveiled in this study suggest that breathing efficiency is affected by ontogenetic stage and local climate adaptation which are significant factors to consider when evaluating the overall vulnerability of intertidal ectotherms to temperature increases. These findings highlight the need to recognize that the evolutionary adaptations of dual-breathing modes acquired by some intertidal ectotherms may be a key mechanism to ameliorate the effects of climate change. Focussing on the organisms that have acquired such mechanisms will greatly improve our understanding on the impact of climate changes on intertidal organisms, populations and communities.

### References

Anger, K. 2001. The biology of decapod crustacean larvae (*Crustacean Issues* 14), AA Balkema Publishers, The Netherlands. hdl:10013/epic.15410.d001

- Bates, D., M. Mächler, B. M. Bolker, and S. C. Walker. 2015. Fitting linear mixed-effects models using lme4. J. Stat. Softw. **67**: 1–48. doi:10.18637/jss.v067.i01
- Benjamini, Y., and Y. Hochberg. 1995. Controlling the false discovery rate: A practical and powerful approach to multiple testing. J. R. Stat. Soc. Ser. B 57: 289–300. doi:10.1111/ j.2517-6161.1995.tb02031.x
- Cannicci, S., D. Burrows, S. Fratini, T. J. Smith, J. Offenberg, and F. Dahdouh-Guebas. 2008. Faunal impact on vegetation structure and ecosystem function in mangrove forests: A review. Aquat. Bot. 89: 186–200. doi:10.1016/j.aquabot. 2008.01.009
- Charmantier, G., L. Giménez, M. Charmantier-Daures, and K. Anger. 2002. Ontogeny of osmoregulauon, physiological plasticity and larval export strategy in the grapsid crab *Chasmagnathus granulata* (Crustacea, Decapoda). Mar. Ecol. Prog. Ser. **229**: 185–194. doi:10.3354/meps229185
- Clarke, A. 2017, Principles of thermal ecology: Temperature, energy and life. Oxford Univ. Press.
- Dahlke, F. T., S. Wohlrab, M. Butzin, and H.-O. Pörtner. 2020. Thermal bottlenecks in the life cycle define climate vulnerability of fish. Science **369**: 65–70. doi:10.1126/science.aaz3658
- Deutsch, C. A., J. J. Tewksbury, R. B. Huey, K. S. Sheldon, C. K. Ghalambor, D. C. Haak, and P. R. Martin. 2008. Impacts of climate warming on terrestrial ectotherms across latitude. Proc. Natl. Acad. Sci. USA **105**: 6668–6672. doi:10. 1073/pnas.0709472105
- Fernandez, M., N. Ruiz-Tagle, S. Cifuentes, H.-O. Portner, and W. Arntz. 2003. Oxygen-dependent asynchrony of embryonic development in embryo masses of brachyuran crabs. Mar. Biol. **142**: 559–565. doi:10.1007/S00227-002-0965-8
- Fratini, S., S. Cannicci, F. Porri, and G. Innocenti. 2019. Revision of the *Parasesarma guttatum* species complex reveals a new pseudocryptic species in south-east African mangroves. Invertebr. Syst. **33**: 208–224. doi:10.1071/IS18028
- Frederich, M., and H. O. Pörtner. 2000. Oxygen limitation of thermal tolerance defined by cardiac and ventilatory performance in spider crab, *Maja squinado*. Am. J. Physiol. Regul. Integr. Comp. Physiol. **279**: R1531–R1538. doi:10. 1152/ajpregu.2000.279.5.r1531
- Fusi, M., F. Giomi, S. Babbini, D. Daffonchio, C. D. Mcquaid, F. Porri, and S. Cannicci. 2015. Thermal specialization across large geographical scales predicts the resilience of mangrove crab populations to global warming. Oikos 24: 784–795. doi:10.1111/oik.01757
- Gunderson, A. R., and J. H. Stillman. 2015. Plasticity in thermal tolerance has limited potential to buffer ectotherms from global warming. Proc. R. Soc. B Biol. Sci. 282: 20150401. doi:10.1098/rspb.2015.0401
- Hamdoun, A., and D. Epel. 2007. Embryo stability and vulnerability in an always changing world. Proc. Natl. Acad. Sci. USA **140**: 1745–1750. doi:10.1073/pnas.0610108104
- Hartnoll, R. G. 2006. Reproductive investment in Brachyura. Hydrobiologia **557**: 31–40. doi:10.1007/s10750-005-9305-6

- Hayashida, H., R. J. Matear, P. G. Strutton, and X. Zhang. 2020. Insights into projected changes in marine heatwaves from a high-resolution ocean circulation model. Nat. Commun. **11**: 4352. doi:10.1038/s41467-020-18241-x
- Huey, R. B., and J. G. Kingsolver. 1989. Evolution of thermal sensitivity of ectotherm performance. Trends Ecol. Evol. 4: 131–135. doi:10.1016/0169-5347(89)90211-5
- Huey, R. B., M. R. Kearney, A. Krockenberger, J. A. M. Holtum, M. Jess, and S. E. Williams. 2012. Predicting organismal vulnerability to climate warming: Roles of behaviour, physiology and adaptation. Philos. Trans. R. Soc. B Biol. Sci. 367: 1665–1679. doi:10.1098/rstb.2012.0005
- IPCC. 2014. Climate change 2014: Synthesis Report. Contribution of Working Groups I, II and II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change.
- Kellermann, V., S. L. Chown, M. F. Schou, I. Aitkenhead, C. Janion-Scheepers, A. Clemson, M. T. Scott, and C. M. Sgrò. 2019. Comparing thermal performance curves across traits: How consistent are they? J. Exp. Biol. **222**: jeb193433. doi: 10.1242/jeb.193433
- Oliver, E. C. J., J. A. Benthuysen, S. Darmaraki, M. G. Donat, A. J. Hobday, N. J. Holbrook, R. W. Schlegel, and A. Sen Gupta. 2021. Marine Heatwaves. Ann. Rev. Mar. Sci. **13**: 313–342. doi:10.1146/annurev-marine-032720-095144
- Padfield, D., H. O'Sullivan, and S. Pawar. 2021. rTPC and nls. Multstart: A new pipeline to fit thermal performance curves in R. Methods Ecol. Evol. **00**: 1–6. doi:10.1111/2041-210X. 13585
- Pinsky, M. L., A. M. Eikeset, D. J. McCauley, J. L. Payne, and J. M. Sunday. 2019. Greater vulnerability to warming of marine versus terrestrial ectotherms. Nature 569: 108–111. doi:10.1038/s41586-019-1132-4
- Pörtner, H. O. 2001. Climate change and temperaturedependent biogeography: Oxygen limitation of thermal tolerance in animals. Naturwissenschaften 88: 137–146. doi:10.1007/s001140100216
- Pörtner, H. O., and A. P. Farrell. 2008. Physiology and climate change. Science 322: 690–692. doi:10.1126/science.1163156
- Pörtner, H. O., and J. Gutt. 2016. Impacts of climate variability and change on (marine) animals: Physiological underpinnings and evolutionary consequences. Integr. Comp. Biol. 56: 31–44. doi:10.1093/icb/icw019
- Przesławski, R., M. Byrne, and C. Mellin. 2015. A review and meta-analysis of the effects of multiple abiotic stressors on marine embryos and larvae. Glob. Chang. Biol. **21**: 2122–2140. doi:10.1111/gcb.12833
- R Core Team. 2019. R: A language and environment for statistical computing.
- Schielzeth, H., and S. Nakagawa. 2013. Nested by design: Model fitting and interpretation in a mixed model era. Methods Ecol. Evol. 4: 14–24. doi:10.1111/j.2041-210x. 2012.00251.x

- Schurmann, H., and J. F. Steffensen. 1992. Lethal oxygen levels at different temperatures and the preferred temperature during hypoxia of the Atlantic cod, *Gadus morhua* L. J. Fish Biol. **41**: 927–394. doi:10.1111/j.1095-8649.1992. tb02720.x
- Shah, A. A., B. A. Gill, and A. C. Encalada. et al 2017. Climate variability predicts thermal limits of aquatic insects across elevation and latitude. Funct. Ecol. **31**: 2118–2127. doi:10. 1111/1365-2435.12906
- Shah, A. A., H. A. Woods, and J. C. Havird. and others 2021. Temperature dependence of metabolic rate in tropical and temperate aquatic insects: Support for the climate variability hypothesis in mayflies but not stoneflies. Glob. Chang. Biol. 27: 297–311. doi:10.1111/gcb.15400
- Shih, H. T., P. K. L. Ng, P. J. F. Davie, C. D. Schubart, M. Türkay, R. Naderloo, D. Jones, and M.-Y. Liu. 2016. Systematics of the family Ocypodidae Rafinesque, 1815 (Crustacea: Brachyura), based on phylogenetic relationships, with a reorganization of subfamily rankings and a review of the taxonomic status of Uca Leach, 1814, Sensu lato and its subgenera. Raffles Bull. Zool. **64**: 139–175.
- Shih, H. T., B. K. K. Chan, and P. K. L. Ng. 2018. *Tubuca alcocki*, a new pseudocryptic species of fiddler crab from the Indian ocean, sister to the southeastern African *T. urvillei* (H. Milne Edwards, 1852) (Crustacea, Decapoda, Brachyura, Ocypodidae). Zookeys **747**: 41–62. doi:10.3897/zookeys. 747.23468
- Simoni, R., S. Cannicci, K. Anger, H. O. Portner, and F. Giomi. 2011. Do amphibious crabs have amphibious eggs? A case study of *Armases miersii*. J. Exp. Mar. Bio. Ecol. **409**: 107– 113. doi:10.1016/j.jembe.2011.08.010
- Small, D. P., P. Calosi, D. Boothroyd, S. Widdicombe, and J. I. Spicer. 2015. Stage-specific changes in physiological and life-history responses to elevated temperature and pCO2 during the larval development of the European lobster homarus gammarus (L.). Physiol. Biochem. Zool. 88: 494– 507. doi:10.1086/682238
- Sokolova, I. M. 2013. Energy-limited tolerance to stress as a conceptual framework to integrate the effects of multiple stressors. Integr. Comp. Biol. **53**: 597–608. doi:10.1093/icb/ict028
- Stevens, G. C. 1989. The latitudinal gradient in geographical range: How so many species coexist in the tropics. Am. Nat. **133**: 240–256.
- Sunday, J. M., A. E. Bates, and N. K. Dulvy. 2011. Global analysis of thermal tolerance and latitude in ectotherms. Proc. R. Soc. B Biol. Sci. 278: 1823–1830. doi:10.1098/rspb.2010. 1295
- Sunday, J. M., A. E. Bates, and N. K. Dulvy. 2012. Thermal tolerance and the global redistribution of animals. Nat. Clim. Chang. **2**: 686–690. doi:10.1038/nclimate1539
- Sunday, J. M., J. M. Bennett, and P. Calosi. and others 2019. Thermal tolerance patterns across latitude and elevation.

Vorsatz et al.

Philos. Trans. R. Soc. B Biol. Sci. **374**: 20190036. doi:10. 1098/rstb.2019.0036

- Verberk, W. C. E. P., J. Overgaard, R. Ern, M. Bayley, T. Wang, L. Boardman, and J. S. Terblanche. 2016. Does oxygen limit thermal tolerance in arthropods? A critical review of current evidence. Comp. Biochem. Physiol. Pt A Mol. Integr. Physiol. **192**: 64–78. doi:10.1016/j.cbpa.2015.10.020
- Vorsatz, L. D., P. Pattrick, and F. Porri. 2021. Fine-scale conditions across mangrove microhabitats and larval ontogeny contributes to the thermal physiology of early stage brachyurans (Crustacea: Decapoda). Conserv. Physiol. **9**: 1– 14. doi:10.1093/conphys/coab010
- Walther, K., K. Anger, and H. O. Pörtner. 2010. Effects of ocean acidification and warming on the larval development of the spider crab *Hyas araneus* from different latitudes (54° vs. 79° N). Mar. Ecol. Prog. Ser. **417**: 159–170. doi:10.3354/meps08807
- Wobbrock, J. O., L. Findlater, D. Gergle, and J. J. Higgins. 2011. The aligned rank transform for nonparametric

factorial analyses using only anova procedures. Proceedings of the SIGCHI conference on human factors in computing systems: 143–146.

### Acknowledgments

FP acknowledges use of infrastructure provided by the South African Institute for Aquatic Biodiversity Research Platform – National Research Foundation of South Africa. SC thanks the Kenya Marine and Fisheries Research Institute for the kind assistance while working at the Gazi Field station. Funds for this research were provided by the SP3- People (Marie Curie) IRSES Project CREC (no. 247514) and the South African Research Chairs Initiative of the Department of Science and Technology and the National Research Foundation (Grant number 64801).

> Submitted 16 July 2021 Revised 08 November 2021 Accepted 10 November 2021